Author's response to reviews

Title: Pseudonocardia carboxydivorans in human cerebrospinal fluid: A case report in a patient with traumatic brain injury

Authors:
AMALIA NAVARRO-MARTÍNEZ (anavarrom@sescam.jccm.es)
NOELIA COROMINAS (n_corominas@hotmail.es)
CARIDAD SAINZ DE BARANDA (csainzd@sescam.jccm.es)
ANGEL ESCUDERO-JIMÉNEZ (aescudero021281@mail.com)
JORGE GALÁN ROS (aescudero021281@gmail.com)
JUAN SÁEZ-NIETO (oficina.informacion@sciii.es)
JAVIER SOLERA (solera53@yahoo.es)

Version: 2 Date: 03 Jun 2017

Author’s response to reviews:

Dear reviewers:

Thank you for your attention. I have edited the following points:

1. The additional files have been removed and legend has been included in the main manuscript.

2. The email addresses of all the authors have been included on the title page.

3. A clean copy, accepting tracked changes has been submitted.

The manuscript has been revised and corrected the order of references.

I apologize for the delay in sending the manuscript.

Best wishes, Amalia Navarro-Martínez.